ORIGINAL ARTICLE



Higher quantiles of sea levels rise faster in Baltic Sea Climate projections

Christian Dieterich¹ · Hagen Radtke²

Received: 23 April 2023 / Accepted: 26 December 2023 © The Author(s) 2024

Abstract

Absolute sea levels in the Baltic Sea will rise under the influence of climate warming, similar to those in the world ocean. For extreme sea levels, there are indications that they will rise even faster than mean sea levels, but that topic is still controversially discussed and existing studies point into different directions. We analyzed a regional climate model ensemble for the Baltic Sea for future sea level changes. We find that the rate of change differs between high sea levels and the average: In the eastern part of the Baltic Sea, the 99th percentile of sea level was predicted to rise faster than the median. In the south-western part, the relation was opposite. Thus, our simulations predict a change not only in the sea level mean, but also in its distribution. This pattern was almost consistent between the individual ensemble members. We investigated the 99th percentile as a proxy for extreme sea levels, since their partially stochastic nature limits the predictive skill of our 20-member ensemble. Our findings imply that adapting coastal protection to mean sea level change only may be regionally insufficient.

Plain Language Summary

We simulated sea level rise in the Baltic Sea with a regional model. We analysed whether high sea levels, as they occur during a storm, rise faster than the mean sea levels. Our model suggests that this happens in the eastern Baltic Sea, while in the south-western part the rise is less fast. This information might be useful for coastal protection against flooding.

1 Introduction

1.1 Mean sea level rise in the Baltic Sea

As the Baltic Sea is a marginal sea, its absolute sea level is controlled by three main factors. The first is the sea level in the world oceans, which propagates into the Baltic Sea via its connection to the North Sea. The second is its low salinity which causes an additional steric component, elevating absolute sea levels specifically in the innermost parts of the sea, where the lowest densities occur. A third major component

Hagen Radtke hagen.radtke@io-warnemuende.de

¹ Department of Research and Development, Swedish Meteorological and Hydrological Institute, 60176 Norrköping, Sweden

² Department of Physical Oceanography and Instrumentation, Leibniz Institute for Baltic Sea Research Warnemünde, 18119 Rostock, Germany is local atmospheric forcing. Wind and air pressure create a dynamic component of the sea level signal. All three drivers, global sea level, Baltic Sea salinity and local atmospheric patterns, are predicted to change in a warming climate. For the period of 1960–2015, the influence of the global mean sea level was largest with about 75% of the variability, but not the only driver, also intensified westerly winds made the sea levels in the northernmost parts of the Baltic Sea rise faster than the global mean (Gräwe et al. 2019). Such trends in the past cannot easily be attributed to climate change though because of the considerable multidecadal variability (Weisse et al., 2021). Regional model projections for future climate scenarios show rising absolute sea levels throughout the Baltic Sea, similar to trends in the North Atlantic(Meier et al. 2022b).

For sea level changes relative to land, the vertical land movement needs to be taken into account, which is, in the Baltic Sea, dominated by glacial isostatic adjustment. The melting of the ice shield after the last glaciation removed weight from the Scandinavian peninsula, which still reacts to this with an uplift of up to 9.6 mm/year (Vestol et al. 2019). The southern coast of the Baltic Sea, in contrast,

C. Dieterich is deceased.

experiences a land subsidence as a compensating effect. This causes mean relative sea levels to rise in the south and fall in the north of the Baltic Sea, as can be seen from tide gauge records dating back to the 19th century. In the recent decades, an acceleration is seen in the long-term trends since the 1960s (Karimi et al. 2021), consistent with the global signal (Dangendorf et al. 2019).

1.2 Changes in sea level extremes

Extreme sea levels in the Baltic Sea are typically caused by storm surges, which can raise the local sea level by up to 3-4 m above the mean (Wolski and Wisnewski 2020). Additionally, the filling state of the Baltic Sea plays a role, westerlies over a period of several days can increase the water volume of the Baltic Sea and lead to a background sea level elevation of ~24 cm, on top of which even less strong cyclones can cause sea level extremes (Weisse and Weidemann 2017). Specifically meteotsunamis can locally produce large and fast local elevations (Pellikka et al. 2020). The Baltic Sea differs from other regions by the fact that the tidal signal is small and only marginally contributes to sea level elevations. While the coincidence of meteorological extremes with high tides is required for other seas, the extreme sea levels in the Baltic Sea are therefore only controlled by meteorological forcing and modified by ice conditions, which can reduce coastal sea level elevations.

There is a controversy in whether extreme sea levels rise or will rise faster than mean sea levels in the Baltic Sea. The quantile regression method, the same as we use in our study, has been applied to observational data before. Barbosa (2008) used it on monthly-mean sea level records at several stations around the Baltic Sea, which had been deseasonalized by subtracting a monthly climatology. She found that specifically in the northernmost part, the higher quantiles were rising systematically faster than the mean. However, these high quantiles refer to entire months with an extraordinarily high average sea level, rather than to individual extreme events. Ribeiro et al. (2014) analyzed daily time series at 7 stations along the Danish and Swedish coast for their extremes. From the quantile regression method, they found that extreme sea levels were rising faster than the mean sea levels at 6 out of these 7 stations, with highest discrepancies (up to 2 mm/year) in the northernmost part. A second method was fitting generalized extreme value distributions. Three out of seven of their time series were significantly better fitted when a trend in the location of the extreme value distribution was assumed, that is, assuming a change in the total distribution of sea levels. Meier et al. (2022), however, argue that the trend, that was clearly identified in the observations, cannot necessarily be attributed to a systematic change in climate, since also climate variability, and specifically a random occurrence of the strongest individual storm surges in the latter part of the time series, could have as well caused the signal. The same is true for the study of Wolski and Wisnewski (2021), who investigated annual maximum surge levels and found a positive trend in 1960–2020 for stations all around the Baltic Sea.

Disentangling whether variability or a systematic trend is the reason for the observed changes is complicated by the limited length of the available time series, and correspondingly the few available data points that contribute to the trend in the higher percentiles. Also, simply adding more stations to the analysis will not necessarily help, since spatial covariance makes them not independent. Model studies could help in answering this question, since they can (a) simulate climate change in an ensemble of model runs with a different random signal, and (b) in the best case help to understand the dynamic cause for the changes in the higher percentiles and attribute it to a change in the forcing.

1.3 Extreme sea levels in Baltic Sea regional climate simulations

Extreme sea levels are influenced and determined by changes in the wind field. Atmospheric ensemble simulations show small changes in the wind field over the Baltic Sea region. Both mean and extreme 10-m wind speeds may change by a few percent, with an increase in winter and a decrease in summer, over the Baltic Sea region, but the signal has large uncertainties as the ensemble spread is much larger than the signal, even in the RCP8.5 scenario (Christensen et al., 2022). Climate simulations have also addressed extreme sea levels directly. An ensemble simulation by Vousdoukas et al. (2016) with 8 different climate scenarios confirmed the findings of Ribeiro et al. (2014): It projected larger trends for extreme sea levels, specifically sea levels with return periods of 5-100 years, relative to mean sea levels in the Baltic Sea. In its northern part, most ensemble members were consistent in predicting this trend in the shape of the sea level distribution. A shortcoming of the study was that the model was a shallow-water model which means that baroclinic effects such as the permanent halocline in the Baltic Sea could not be represented. It was directly driven by the atmospheric output of global climate models of the Coupled Model Intercomparison Project 5 (CMIP5).

Another study by Paprotny et al. (2016) used atmospheric downscalings of climate projections to drive a similar ocean model. Rather than applying the atmospheric forcing from a global simulation directly, they used a regional atmospheric model (RCA4 by the Swedish Meteorological and Hydrological Institute) which had been driven by lateral boundary conditions from a global climate model, ICHEC-EC-EARTH. Benefits are a finer resolution of atmospheric dynamics in the target region, specifically the use of a more realistic topography and land mask. Their study showed the opposite result, declining storm surges in the Baltic Sea, in both an RCP4.5 and RCP8.5 climate scenario. A shortcoming of this study is the use of a single realization of each future climate scenario, instead of a model ensemble.

Our study investigates the same question with a dynamic regional downscaling simulation: Atmosphere and ocean are two-way coupled in our setup, which adds the ability of the system to represent interactions between ocean, ice and atmosphere. The ocean is represented by a full 3-dimensional model which includes baroclinic effects. Also, we use an ensemble of several climate scenarios driven by different global models to answer the question of changing extremes.

1.4 Outline

The article is structured as follows: In Sect. 2, we will describe the model system that we used and the statistical methods that we used for evaluating the model results. In Sect. 3, we describe the observational data we used, specifically for model validation. Section 4 shows our results, starting with a model skill assessment for the given purpose and then presenting our results on the changes in 99th percentile vs. 50th percentile sea levels. Section 5 ends with conclusions.

2 Materials and methods

2.1 Regional climate model simulations

We use a regional coupled model to do a downscaling of global climate simulations, which are used as boundary conditions. Our model system consists of an atmospheric and an ice-ocean component. The atmospheric model is the Rossby Centre regional atmospheric model (RCA4, Kjell-ström et al. 2016) in a resolution of 0.22° in the horizontal and 40 vertical layers. It is covering the EURO-CORDEX domain, i.e., a region including almost the entire European continent and parts of northern Africa. The ice-ocean model

is NEMO-Nordic (Hordoir et al. 2019), which is applied for the North Sea and Baltic Sea, with a horizontal resolution of approximately 2 nautical miles and 56 vertical z-levels. More details on the model setup can be found in Dieterich et al. (2019a, b).

An ensemble of runs is performed by this regional model, taking different global simulations as lateral forcing. The forcing data (see Table 1) include one hindcast simulation (the ERA-40 reanalysis) and eight climate models that were used in the coupled model intercomparison project CMIP5. Each of these models contributes one reference simulation for the historical period and two to three climate scenarios, which differ in their radiative forcing. Simulations for three reactive concentration pathways (RCPs) were chosen from the list of available model experiments, the RCP2.6, RCP4.5 and RCP8.5 scenarios, which represent an optimistic, a moderate and a strongly pessimistic development of atmospheric greenhouse gas concentrations during the 21st century. For each climate scenario, a transient simulation was performed starting in 1961-2005 with the historical reference and continued to including 2099 with the respective climate scenario. In total, this amounts to 20 transient simulations in addition to the 1961-2009 downscaling of the ERA-40 hindcast.

2.2 Contributions to the sea level signal

The local sea level change simulated by the model consists of different contributions. Global sea level rise enters the model via the open boundary conditions, since it is partly simulated by the global models. Our barotropic boundary conditions, prescribed as sea levels, are a sum of two signals: The tidal signal is defined by 11 constituents from a tidal model (Egbert et al. 2010), a non-tidal signal is added to these from a global simulation: In the hindcast simulation, we use the ORAS4 reanalysis (Balmaseda et al. 2013), in the climate scenarios we extract monthly mean sea level time series from the global models. These capture the thermal expansion of the ocean, a local steric contribution as well

Global model	Number of simula- tions	Scenarios	Years
ERA-40	1	Reanalysis	1961–2009
CANESM2	2	Historical + RCP4.5/RCP8.5	1961-2099
CNRM-CM5	2	Historical + RCP4.5/RCP8.5	1961-2099
EC-EARTH	3	Historical + RCP2.6/RCP4.5/RCP8.5	1961-2099
GFDL-ESM2M	3	Historical + RCP2.6/RCP4.5/RCP8.5	1961-2099
IPSL-CM5A	2	Historical+RCP4.5/RCP8.5	1961-2099
MIROC5	2	Historical+RCP4.5/RCP8.5	1961-2099
MPI-ESM-LR	3	Historical+RCP2.6/RCP4.5/RCP8.5	1961-2099
NORESM1-M	3	Historical + RCP2.6/RCP4.5/RCP8.5	1961–2099

Table 1 Boundary data usedfor the 20 + 1 simulations in themodel ensemble

as dynamic changes induced by changes in the wind signal or the inverse barometric effect. These dynamic effects are also covered in the regional model. Missing in the model is the effect of the melting of inland ice, as well as the vertical land movement, which needs to be taken into account when considering relative sea level rise. These contributions were therefore added on top, after the simulations were finished. For the meltwater signal, we used global estimates from the IPCC fifth assessment report (IPCC, 2013), thereby neglecting regional effects of a change of the gravity field by the melting inland ice. Only in the reanalysis run, this mass signal did not have to be added, since it was contained in the global model already. For the vertical land movement, which is pronounced in the Baltic Sea area, we used estimates from the NKG2016LU model (Vestøl et al. 2019). Adding these contributions to the results only means that we have to assume a linear combination of the signals. A possible non-linear interaction of drivers, such as a change in the dynamic coastal sea level signal after a meltwaterinduced global sea level rise, cannot be taken into account by our approach.

Daily maxima of hourly mean sea levels were stored by the model in all coastal grid cells, in order to assess the changes in sea level extremes. We only analyze the model output from 1971 onward, the first ten years of the simulation are considered as spin-up period.

2.3 The quantile regression method

Rather than looking at sea level extremes in the model, we analyse the development of the 99th percentile of sea levels as a proxy for changes in extremes. The reason for this is that climate-induced trends in sea level extremes are difficult to assess, since the occurrence of extremes has a stochastic component, which is driven by weather rather than climate. In reality as well as in model-generated sea level time series, this random component occurs on top of a systematic signal. In numerical models, it is in principle possible to eliminate this random part of the signal by running a large ensemble of simulations, each with perturbations, such that the stochastic component will be independent between the runs and disappear in the ensemble mean. The disadvantages of this approach from a regional modelling perspective are on the one hand the numerical effort and on the other hand its dependency on the existence of large enough global-model ensembles.It is, for our purpose, not helpful to generate regional ensembles by adding perturbations at the regional scale. This is because model runs driven by the same global boundary conditions, even if they may develop different internal variablity, would still share the stochastic components from the open-boundary forcing and thus not be independent realizations of a climate state. Treating them statistically as independent members might thus lead to overconfidence and the misidentification of stochastic contributions as systematic trends.

While from a coastal engineering perspective, extreme sea levels are of interest, which means long return periods, their trend may show a strong stochastic component even in a model ensemble due to the limited number of events. It is therefore necessary to look at less extreme values which occur more frequently and give better statistics. A necessary condition for the model study to make sense would be that the regional model is able to reproduce the observed trends when being forced with realistic (hindcast) past boundary conditions. We will demonstrate that this is true for trends in the 99th percentile of sea levels in our model system.

We therefore use the quantile regression method (Koenker and Hallock 2001) to estimate trends in different percentiles of the sea level time series. It works similar to linear regression and estimates a linear fit through data points in a time series. In linear regression, the sum of the squared errors between model and data is minimised, this quadratic penalty function makes the trend line estimate the mean value. A penalty function using the absolute errors will, if minimized, make the trend line estimate the median (the 50th percentile). Lines that estimate trends in higher and lower percentiles can be obtained by using a suitably asymmetric penalty function. We use the software R (Core Team 2021) and the package quantreg (Koenker 2021) for the analysis and fit the 1st, 50th and the 99th percentile of the sea level time series.

2.4 Assessing model skill

Model skill is assessed by a comparison to selected tide gauge records in the Baltic Sea. The data were extracted from the GESLA database (Haigh et al. 2021) for the stations listed in Table 2 and were aggregated in the same way as the model data (daily maxima of hourly means). Observations were compared to the results of the ERA-40 downscaling simulation in terms of correlation and variance for an assessment of the overall model skill, and multidecadal trends in the percentiles of the sea level time series were analyzed to check the credibility of the model in predicting these percentile trends given the assumption that the boundary data are realistic. To compare these percentile trends, model data were extracted from the closest coastal grid points to the station locations in Table 2. At those points in time where observations had gaps, the model data were also set to missing values. Then, the trends for the three percentiles were calculated over different multidecadal periods and the modelled trends compared to the observed ones.

Table 2 Selected locations	s for sea	level	observations
----------------------------	-----------	-------	--------------

Station name	Latitude (°N)	Longitude (°E)
Degerby	60.03190	20.384800
Furuogrund	64.91580	21.230600
Göteborg	57.68470	11.790600
Hanko	59.82290	22.976601
Helsinki	60.15360	24.956200
Klagshamn	55.52220	12.893600
Kungsholmsfort	56.10530	15.589400
Marviken	58.55360	16.837200
Raahe	64.66590	24.407101
Ratan	63.98610	20.895000
Rauma	61.13350	21.425800
Smögen	58.35360	11.217800
Spikarna	62.36330	17.531100
Travemünde	53.95802	10.872182
Turku	60.42830	22.100500
Visby	57.63920	18.284400

3 Results

3.1 Overall model skill on simulating sea level

To analyse whether the model is capable of reproducing observed sea levels, we compare observations at GESLA stations in the Baltic Sea to model results at the closest coastal points. We compare the daily maxima of hourly means. Figure 1a shows that the agreement between model



and observations depends on the region in the Baltic Sea: The model explains more than 70% of the variance east of 15°E, with the exception of the Raahe station where agreement is lower. Correlations are slightly lower but show the same spatial pattern if deseasonalized data are used. The low skill in the western part can be explained by a shortcoming in the choice of the boundary conditions for the model ensemble, see Sect. 4.1. The variance in the daily maxima of the sea levels is systematically underestimated in the model, see Fig. 1b. The underestimation however, only amounts to around 10-20% of the signal, and the spatial variability in the temporal variance is captured by the model. Exceptions are the westernmost stations Travemünde, Smögen and Göteborg where the underestimation is pronounced. The model correctly finds that the distribution of the daily-maxima sea levels is right-skewed for all stations. Its median is below the mean for all stations, and it's 97.5th percentile has a larger distance to the mean than the 2.5th percentile.

3.2 Model skill in simulating past changes in sea level percentiles

The trends in the 1st, 50th and 99th percentile of sea level were calculated for all six possible multidecadal periods with a starting or ending date of January 1st in 1970, 1980, 1990, 2000 or 2010. As these were six periods (partially overlapping), we got a vector of 18 trend values for the observed sea levels and another one for the reconstructed sea levels by the ERA-40 downscaling. Figure 2a and b illustrates how some modelled and observed trends



Fig. 1 a Squared correlation coefficient between modelled sea level time series and GESLA observation data. **b** Observed (blue) vs. modelled (red) spread in the daily maxima. Vertical bars denote the inter-

val that contains 95% of the values. Horizontal ticks indicate the location of the median relative to the mean



Fig. 2 Ability of the model to reproduce interdecadal trends in the sea level quantiles. (**a-b**) Trends at station Degerby / Kungsholmsfort for different two-decadal periods. The black and grey curves show observations and model results, respectively. The solid lines indicate the

match for two stations, Degerby (with a good match) and Kungsholmfort (with a slightly worse match). Figure 2c shows the squared correlation coefficient between the observed and modelled trend vectors. While in the western part of the Baltic Sea (west of 15°E), the fraction of the variance in the observed trends that is explained by the model is less than half, these values improve in the eastern part where the model explains the majority of the variance in the multidecadal variations, with the exception of the Spikarna station. This gives us confidence that for this region, the downscaling model has skill in predicting the multidecadal trends in the selected sea level quantiles, given that the boundary data are realistic.

multidecadal trends in the observations and the dashed lines the corresponding trends in the model reconstruction. c Squared correlation coefficient between observed and simulated multidecadal trends in the 50th and 99th percentile for different stations

3.3 Projected changes in median sea level

For the 50th percentile of sea level, that is, the median value of daily maxima of hourly means, our model simulates a bipolar pattern in sea level change: Relative sea levels are predicted to rise with several millimeters per year at the southern part of the Baltic Sea, while we find a relative sea level drop of again several millimeters in Scandinavia (Fig. 3). This is a well-known pattern also observed today and a result of the pronounced vertical land movement caused by glacial isostatic adjustment. Sea level rise shifts the zero-line of mean sea level change northward, in our model scenarios for RCP2.6 and RCP4.5 it crosses the Swedish coast at the öland and Gotland islands and the Estonian coast at the island of Saarema. In the more



Fig. 3 Ensemble mean of the linear trend in the 50th percentile of sea level for 1971–2099

pessimistic RCP8.5 scenario, the isoline is shifted slightly northward and these islands experience a rise in relative sea level around their entire coast. Our model results are consistent with tide-gauge-based reconstuctions for the period 1915–2014 which show a zero-trend-line a bit south of these mentioned islands (Madsen et al. 2019).

3.4 Projected changes in higher sea level quantiles

The 99th percentile shows similar, but not identical, trends as the 50th percentile. When looking at the ensemble mean of the difference between them (Fig. 4), the regional pattern is similar in all reactive concentration pathways: In the south-western part of the Baltic Sea, we see the 99th percentile sea levels rising more slowly than the 50th percentile, while in the northern and eastern part, extreme sea levels rise faster. In the Bothnian Bay in the very north, the signal depends on the chosen climate scenario, with faster rising 99th percentile, consistent with the adjacent Bothnian Sea, in the most pessimistic RCP8.5 scenario, and with 99th percentile sea levels rising more slowly in the other two scenarios. The magnitude of the relatively faster growth of the 99th percentile differs between the scenarios, with values below 0.25 mm/year in the optimistic and intermediate scenarios, but under RCP8.5 up to 0.75 mm/year are reached in the Gulf of Riga and Gulf of Finland. In the south-western part, the 99th percentile rise is slowed down by -0.25 to - 0.5 mm/year in all climate scenarios.

Looking at the individual models rather than the ensemble mean (Fig. 5), we see that these trends are often consistent between the model runs forced with the different climate scenarios. We consider four stations, Travemünde in the southwestern Baltic Sea,Riga in the Gulf of Riga and Helsinki in the Gulf of Finland as reperesentatives of the eastern Baltic Sea region where the rise signal is most pronounced, and Raahe in the Bothnian Bay, the northernmost basin. For Travemünde, the agreement is largest, with 19 out of 20 scenario runs showing a decreasing trend in the 99th percentile minus the median, and only one scenario showing the opposite trend. For the two eastern stations Riga and Helsinki, we see that the agreement between the models rises with the radiative forcing: 2 out of 4 (6 out of 8, 7–8 out of 8) models agree on the positive sign of the trend for the RCP2.6 (RCP4.5, RCP8.5) scenarios. For the station Raahe in the Bothnian Bay, models show agree on a negative trend in the RCP2.6 scenario but increasingly disagree on the direction of the trend in the warmer model scenarios.

Our last step is to investigate whether the trends are robust against interdecadal variablility. We want to check whether it is possible that a single strong storm could turn the trend around when it occurs in an early or in a late decade. To do so, we omit single decades from the modelled time series and repeat the trend calculation. The results are shown in Fig. 6. It can be seen that omitting single decades would change these trends by ~ 0.25 mm/year at most. Only few points occur in that graph in the second or fourth quadrant, where they indicate a switching of the sign in the signal. Only three situations occur where a positive trend turns negative by removing a decade from the time series, all of them for the Raahe station, so the positive trends at the Riga and Helsinki stations are systematic in a sense that they are not determined by extreme events in an individual decade of the simulation.

4 Discussion

4.1 Model applicability

The comparison of the modelled sea levels to observations shows that the model has skill in reproducing both daily maxima and also decadal trends of different quantiles of sea levels. The agreement is poor in the western Baltic Sea. This is probably due to the coarse resolution that does not resolve the topography in the Danish Straits, and due to the choice of the sea surface height boundary conditions at the borders between the North Sea and the Atlantic Ocean. Dieterich et al. (2019) showed that extreme sea levels in the



Fig. 4 Ensemble mean of the difference of linear trends, 99th minus 50th percentile, of sea level for 1971–2099



Fig. 5 The difference between 99th and 50th percentile trends for individual ensemble members for selected stations

model strongly depend on the specific choice, and Gröger et al. (2019) showed that the present choice of the boundary conditions causes unrealistic circulation patterns in the North Sea, which will likely affect the transition region in the Baltic Sea as well.

For the central and eastern part of the Baltic Sea, our model outperforms the barotropic models used in previous studies. Explained variance in the daily maxima is above 0.7, while Paprotny et al. (2016) states values around 0.6 for Northern European stations. Our root-mean-squared error values are 0.13 m as average over all Baltic Sea stations, here both Vousdoukas et al. (2016) and Paprotny et al. (2016) give a value of 0.15 m.

Our regional model has shown that it reproduces past multidecadal trends in the sea level quantiles, at least the majority of their variance, when driven by realistic historic atmospheric boundary conditions. So, if the global models succeed in providing realistic projections of the atmospheric circulation over northern Europe in a changed climate, our regional model system should also be able to translate these to realistic future sea level distributions.

4.2 Discussion of model results

The results of our simulations suggest that in the eastern part of the Baltic Sea, climate change will cause a systematically increasing trend in the 99th percentile sea levels relative to the median sea level. This result of our simulations is not significant though; the use of different global climate models reflects our uncertainty in the global response to anthropogenically altered radiative forcing, and this uncertainty propagates into our projections for regional changes. Arriving at 95% confidence would thus only have been possible





with a larger ensemble, or an agreement of all eight of the ensemble members for each scenario.

The results also suggest an opposite trend (declining 99th percentiles relative to the median) in the south-western part of the Baltic Sea. Even if the model agreement is even better here, this result is still less certain, since the model shows larger deviations from observational data in this region.

We did not look into higher percentiles, such as return periods of 100 years, since we could not prove from a comparison to observational data that our model has skill in predicting trends in those. The reason is that the larger the percentile, the more observations are needed to detect a systematic trend against variability. A transferability of our results to higher return periods therefore depends on the question whether these extreme sea levels have a similar generation mechanism as the 99th percentile sea levels for which our model suggested the change.

Seen in the light of the existing literature, our results give some support to the hypothesis of Ribeiro et al. (2014) that observed increases of extreme sea levels relative to the mean show a systematic effect or at least have a systematic component. There is a regional mismatch, though, between our study and those of Barbosa (2008) and Ribeiro (2014): The observed extreme sea level rises were maximal in the northermost part of the Baltic Sea, where our model shows such a signal only for the RCP8.5 scenario under much warmer conditions than at present. Our climate simulations are a complementary method to observations: By their nature, they differ in the random part of the signal. As they show agreement beyond what would be expected in a random ensemble, this means the effect must be systematic. It is therefore likely that it may have occurred already in the past few decades which were influenced by climate warming. For the future scenarios, our study agrees with the projections of Vousdoukas et al. (2016) and questions the model results by Paprotny et al. (2016) which rely on a single realization of each future climate scenario.

5 Conclusions

Our study used a coupled regional atmosphere-ice-ocean model to do a downscaling of 20 CMIP5 global climate simulations for the North Sea and the Baltic Sea. We specifically looked into high sea levels (the 99th percentile of daily maxima) and found that they exhibit a change relative to the median. They rise faster in the eastern parts of the Baltic Sea and less fast in the south-western parts. The signal is almost consistent between the different climate models and increases in magnitude with the radiative forcing, even if this trend in the individual models is not linear with climate forcing.

Our model results are reliable for the Baltic Sea east of 15°E because here modelled and observed sea levels correlate well. Also we could show that our model was capable of reconstructing the changes in the 99th sea level quantiles that occurred in the past decades. The modelled trends are robust against removing single decades from the simulation.

The implications are that for coastal protection, it may not be sufficient to adapt the measures to changes in mean sea level, since the extremes could rise more rapidly. Fortunately, for large parts of the Baltic Sea, the land uplift due to glacial isostatic adjustment will still lead to a decline in relative sea levels. For the south-western Baltic Sea, where a land subsidence will add to the rise in absolute sea levels and increase its impact, the model predicts a slower rise of the extremes, but due to the poor model performance in this region, this remains uncertain.

The most relevant question that arises for future research is the attribution of the trends in the extremes. The explanation cannot be too simple, as the effect is opposite in single ensemble members. For example, the hypothesis that just the reduced ice cover leads to stronger sea level elevations in the eastern parts cannot give the full explanation, since a reduction of the ice cover also happens in those climate scenarios that show the opposite trend. A first look into changes in storminess and in the direction of strong winds also gave no clear indication of a difference between the outlier model with a negative trend and the other ensemble members. So, the physical mechanism remains an open question. Our existing and available data set, which contains the full hydrodynamic results of the model ensemble, should contain the answer. Investigating this and understanding what can cause an increased rise of sea level extremes might be of global relevance, since the yet unknown physical mechanism may as well act in other coastal seas and increase a flooding risk.

Author contributions CD prepared and performed the model simulations, did the postprocessing of the model results, and provided the idea to analyze them for extremes. HR performed the quantile regression, the comparison to observational data, and the writing of the manuscript.

Funding Open Access funding enabled and organized by Projekt DEAL. The research presented in this study is part of the Baltic Earth programme (Earth System Science for the Baltic Sea region; seehttp:// www.baltic.earth, last access: 28 January 2023). The coupled RCA4-NEMO experiments have been performed on the Linux clusters Krypton, Bi and Triolith, all operated by the NSChttp://www.nsc.liu.se/(last access: 26 January 2023). Resources on the Linux cluster Triolith have been made available by Swedish National Infrastructure for Computing (SNIC) grant no. 002/12–25 "Regional climate modeling for the North Sea and Baltic Sea regions". This part of the simulations was performed on resources provided by SNIC at the National Supercomputer Centre (NSC) in Sweden.

This research has been supported by the Swedish Civil Contingencies Agency, MSB (grant no. 2015–3631, HazardSupport), the Länsförsäkringars Forskningsfond (grant no. P02/12, Future flooding risks at the Swedish Coast: Extreme situations in present and future climate), the Swedish Research Council Formas (grant no. 2017–01949, Clime-Marine), and the Swedish National Space Board, Rymdstyrelsen (grant no. 172/13, Assimilating SLA and SST in an operational ocean forecasting model for the North Sea and Baltic Sea using satellite observations and different methodologies). **Data availability** The modelled sea levels at grid points adjacent to the coast are available online under https://doi.org/10.12754/data-2023-0002. The full set of model results is available on request.

The sea level observations can be downloaded from https://www.gesla.org.

The code for the ocean model that is used in RCA4-NEMO is available at https://doi.org/10.5281/zenodo.2643477 (Dieterich and NEMO Team, 2019).

Declarations

Conflict of interest The authors have no relevant financial or non-financial interests to disclose.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Balmaseda MA, Mogensen K, Weaver AT (2013) Evaluation of the ECMWF ocean reanalysis system ORAS4. Q J R Meteorol Soc 139:1132–1161. https://doi.org/10.1002/qj.2063
- Barbosa SM (2008) Quantile trends in Baltic Sea level. Geophys Res Lett 35:2008GL035182. https://doi.org/10.1029/2008GL035182
- Core Team R (2021) R: a language and environment for statistical computing. R Foundation for Statistical Computing, Vienna
- Dangendorf S, Hay C, Calafat FM, Marcos M, Piecuch CG, Berk K, Jensen J (2019) Persistent acceleration in global sea-level rise since the 1960s. Nat Clim Chang 9:705–710. https://doi.org/10. 1038/s41558-019-0531-8
- Dieterich C, Gröger M, Arneborg L, Andersson HC (2019a) Extreme sea levels in the Baltic Sea under climate change scenarios – Part 1: model validation and sensitivity. Ocean Sci 15:1399–1418. https://doi.org/10.5194/os-15-1399-2019
- Dieterich C, Wang S, Schimanke S, Gröger M, Klein B, Hordoir R, Samuelsson P, Liu Y, Axell L, Höglund A, Meier HEM (2019b) Surf heat budget over North Sea Clim Change Simulations. Atmosphere 10:272. https://doi.org/10.3390/atmos10050272
- Egbert GD, Erofeeva SY, Ray RD (2010) Assimilation of altimetry data for nonlinear shallow-water tides: quarter-diurnal tides of the Northwest European shelf. Cont Shelf Res 30:668–679
- Gräwe U, Klingbeil K, Kelln J, Dangendorf S (2019) Decomposing mean sea level rise in a semi-enclosed basin, the Baltic Sea. J Clim 32(11):3089–3108. https://doi.org/10.1175/JCLI-D-18-0174.1
- Haigh ID, Marcos M, Talke SA, Woodworth PL, Hunter JR, Haugh BS, Arns A, Bradshaw E, Thompson P (2021) GESLA version 3: a major update to the global higher-frequency sea-level dataset. Geosci Data J. https://doi.org/10.1002/gdj3.174
- Hordoir R, Axell L, Höglund A, Dieterich C, Fransner F, Gröger M, Liu Y, Pemberton P, Schimanke S, Andersson H, Ljungemyr P, Nygren P, Falahat S, Nord A, Jönsson A, Lake I, Döös K, Hieronymus M, Dietze H, Löptien U, Kuznetsov I, Westerlund A, Tuomi L, Haapala J (2019) Nemo-nordic 1.0: a NEMO-based

ocean model for the Baltic and North seas – research and operational applications. Geosci Model Dev 12:363–386. https://doi. org/10.5194/gmd-12-363-2019

- Karimi AA, Bagherbandi M, Horemuz M (2021) Multidecadal Sea Level variability in the Baltic Sea and its impact on Acceleration Estimations. Front Mar Sci 8:702512. https://doi.org/10.3389/ fmars.2021.702512
- Kjellström E, Bärring L, Nikulin G, Nilsson C, Persson G, Strandberg G (2016) Production and use of regional climate model projections – a Swedish perspective on building climate services. Clim Serv 2–3:15–29. https://doi.org/10.1016/j.cliser.2016.06.004
- Koenker R (2021) quantreg: Quantile regression. https://CRAN.R-proje ct.org/package=quantreg. Accessed 29 Jan 2024
- Koenker R, Hallock KF (2001) Quantile regression. J Econ Perspect 15:143–156. https://doi.org/10.1257/jep.15.4.143
- Madsen KS, Høyer JL, Suursaar Ü, She J, Knudsen P (2019) Sea Level trends and variability of the Baltic Sea from 2D statistical reconstruction and altimetry. Front Earth Sci 7:243
- Meier HM, Kniebusch M, Dieterich C, Gröger M, Zorita E, Elmgren R, Myrberg K, Ahola MP, Bartosova A, Bonsdorff E (2022) Climate change in the Baltic Sea region: a summary. Earth Syst Dyn 13:457–593
- Paprotny D, Morales-Nápoles O, Nikulin G (2016) Extreme sea levels under present and future climate: a pan-European database. E3S Web Conf 7(02001):02001. https://doi.org/10.1051/e3sconf/ 20160702001

- Ribeiro A, Barbosa SM, Scotto MG, Donner RV (2014) Changes in extreme sea-levels in the Baltic Sea. Tellus A Dyn Meteorol Oceanogr 66:20921. https://doi.org/10.3402/tellusa.v66.20921
- Vestøl O, Ågren J, Steffen H, Kierulf H, Tarasov L (2019), NKG2016LU: a new land uplift model for Fennoscandia and the Baltic Region. J Geodesy 93(9):1759–1779. https://doi.org/10. 1007/s00190-019-01280-8
- Vousdoukas MI, Voukouvalas E, Annunziato A, Giardino A, Feyen L (2016) Projections of extreme Storm surge levels along Europe. Clim Dyn 47:3171–3190. https://doi.org/10.1007/ s00382-016-3019-5
- Wolski T, Wiśniewski B (2020) Geographical diversity in the occurrence of extreme sea levels on the coasts of the Baltic Sea. J Sea Res 159:101890
- Wolski T, Wiśniewski B (2021) Characteristics and long-term variability of occurrences of Storm surges in the Baltic Sea. Atmosphere 12:1679

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.